Comment:
The authors must be appreciated for a very well written articles. However, few corrections are required as described below. 1. Spellings: There multiple instances where both North American and British spellings are used throughout the document. Kindly use one type of spelling throughout the manuscript. 2. Grammar and Punctuation: Needs to be published. 3. Kindly elaborate criteria for palliative care, and also exactly what palliative care was given to the alcohol related liver disease.

Response:
We appreciate the reviewer’s time and effort for their insightful comments. Here are the responses:

1. We reformatted the manuscript to include american spellings only.
2. Grammatical and punctuation errors were corrected.
3. Palliative care consultation was identified using the ICD codes (ICD 9: V66.7, ICD 10: Z51.5) (The statement has been added to the methods section).